**Author’s response to reviews**

**Title:** Association between residual teeth number in later life and incidence of dementia: A systematic review and meta-analysis

**Authors:**

Bumjo Oh (bumjo.oh@gmail.com)

Dong-Hun Han (dhhan73@gmail.com)

Kyu-Tae Han (kthan.phd@gmail.com)

Xibei Liu (xibeil@deptofmed.arizona.edu)

Johnson Ukken (jukken@med.unr.edu)

Carina Chang (carinac@med.unr.edu)

Kiki Dounis (kdounis@gmail.com)

Ji Yoo (ji.yoo@unlv.edu)

**Version:** 2 **Date:** 19 Jan 2018

**Author’s response to reviews:**

1. We rephrased methods section in pages 4 and 5 to reduce overlap with other previously published works.

2. Email addresses for each author were added in title page.

3. Funding information was removed from the Acknowledgements and was included in the Funding section instead.

4. New data availability statement was taken from one of suggested forms: The dataset generated and/or analyzed during the current study are not publicly available.....

5. A list of all the abbreviations was added in the manuscript just before the Declarations section.

6. A section of additional files was added after the References/Figure legends.

7. Response letter was removed.

8. Revised manuscript was uploaded as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All
tables/figures/additional files (supplementary) files were also cleaned versions and remained as separate files.